Author's response to reviews

Title: Pregnancy Associated Breast Cancer and Increased Risk of Pregnancy-Associated Recurrence

Authors:

Freya Schnabel (writejenn@gmail.com)
Jessica Billig (Jessica.billig@nyumc.org)
Arielle Cimeno (arielle.cimeno@nyumc.org)
Jennifer Chun (jennifer.chun@nyumc.org)

Version: 4 Date: 5 April 2012

Author's response to reviews: see over
April 5, 2012

Dear Editorial Board of the Journal of Medical Case Reports,

Thank you for reviewing our manuscript entitled, “Pregnancy-Associated Breast Cancer and Increased Risk of Pregnancy-Associated Recurrence.”

We have changed the formatting to conform to the journal’s style.

- Kindly address the following: “Please remove dates e.g. 2011 in the case presentation and used phrases like '...was presented to our department 5 months/years ago...' etc.”

✓ We have made these changes in the case presentation. “in 1997” was changed to “first presented to our department fifteen years ago”; “In January 2007” was changed to “Five years ago” and “the patient returned in July 2007” was changed to “the patient returned six months later”.

This is the first report that details a phenomenon of a pregnancy-associated breast cancer recurrence during a subsequent pregnancy. We believe that this case raises the importance of close monitoring and counseling regarding future pregnancies and risks of breast cancer recurrence in this subset of patients. This report has been reviewed by all authors and there is no conflict of interest or financial interest to disclose. This case report has not been published and we hope you will reconsider our revised manuscript for publication in your journal. Thank you for this opportunity.

Sincerely,

Freya Schnabel

Freya Schnabel, MD
Director of Breast Surgery
NYU Clinical Cancer Center
160 East 34th Street
New York, NY 10016
Tel: 212-731-5367